Author's response to reviews

Title: Associations between Impulsivity, Aggression, and Suicide in Chinese College Students

Authors:

lin wang (466373279@qq.com)
lu chen (chenlu900526@gmail.com)
zheng xue qiao (hlisxsysxh@163.com)
xiao hui qiu (hlisxsysxh@163.com)
xiu xian yang (125020874@qq.com)
bing bai (baibing2910@163.com)
Jia Run Yang (1287991708@qq.com)

Version: 2 Date: 9 September 2013

Author's response to reviews: see over
Dear Editor,

According to the Editor's comments, we have carefully revised our paper. Please see the enclosed detailed reply for our concrete modifications.

Here below is our description on revision according to the Editor’ comments.

List of Actions

1. We have added the statement the Methods section whether written informed consent for participation in the study was obtained from participants on the 1th line of p.10.

2. We have added the “Financial competing interests” on the p.20.

3. We have added the “Authors’ contributions” on the p.20.

Thank you and all the kind advice.

Sincerely yours,

Lin Wang.